Author’s response to reviews

Title: Mesenteric panniculitis of the sigmoid colon: a case report and review of the literature

Authors:

Angel I Popkharitov M.D;PhD (popkharitov@abv.bg)
Georgi N Chomov M.D. (g_shomov@abv.bg)

Version: 5 Date: 25 August 2007

Author’s response to reviews:

MEDICAL FACULTY, THRACIAN UNIVERSITY,
Department of Surgery
Tel: (+359) 42 664450, Fax: (+359) 42 600 710
E-mail: popkharitov@abv.bg

22.08.07, Stara Zagora

To: The Editor-in-Chief of
The Journal of Medical Case Reports

Dear Sir,

Enclosed, please, find our revised manuscript MS: 4085700561336578 entitled Mesenteric panniculitis of the sigmoid colon: a case report and review in the literature. We, the authors of the article, Angel Popkharitov and Georgi Chomov, verify that no substantial part of the paper has been or will be published elsewhere. We attest to the fact that all authors listed on the title page have read the manuscript, attest to the validity and legitimacy of the data and its interpretation, and agree to its submission to Journal of Medical Case Reports.

We, the authors of this manuscript declare that they have no competing interests.

We observed and performed the recommendations of the reviewer’s reports:
- The manuscript was re-written and edited according to reviewer’s report;
- We made a more profound research on relevant literature in the Medline and cited recent published articles concerning this entity. We reported the publication by Dr Theresa Emory et al. in the discussion.
- The article was checked by grammatical errors, corrected and edited by native English speaker Dr Theresa Emory, pathologist in Mayo Clinic, USA, for which we express our gratitude in acknowledgment section.
- We changed our mind for radical surgery and added in Conclusion section
anti-inflammatory and immunosuppressive therapy.

We hope that the information and the results of our work will be of interest to the readers of The Journal of Medical Case Reports and will be accepted for publication in the journal.

Sincerely Yours,

Angel Popkharitov, MD, PhD Georgi Chomov, MD
Thracian University, Medical Faculty District Hospital
Dept. of Surgery Dept. of Pathology
11 Armeiska Str., 11 Armeiska Str
6000 Stara Zagora, Bulgaria

tel.: ++ 359 42 644 450 g_shomov@abv.bg
fax: ++359 42 600 710
popkharitov@abv.bg